Author’s response to reviews

Title: The effect of CBT and its modifications for relapse prevention in major depressive disorder: A systematic review and meta-analysis

Authors:
Zuojie Zhang (zuojiezhang@163.com)
Lingli Zhang (zhlingli@sina.com)
Guorong Zhang (essie09@163.com)
Jianing Jin (785forever@163.com)
Zhenyang Zheng (zhenyangzheng@qq.com)

Version: 3 Date: 13 Jan 2018

Author’s response to reviews:

Dear Editors and Reviewers:

Thank you for your letter of January 11, 2018 and for suggestions concerning our manuscript. We are very pleased to learn that our manuscript is acceptable for publication in BMC Psychiatry with minor revision. We quite appreciate your favorite consideration and constructive comments. Now we have revised the paper point-by-point exactly according to this comment.

To Editor

1.  Comment:

Please use Authors' initials in the "Authors' contributions" section instead of their full names.

2.  Response:

Thank you for your advice. Authors' initials were used in the "Authors' contributions" section instead of their full names in the revised manuscript.

Thanks very much for your kind work and consideration on publication of our paper. On behalf of my co-authors, we would like to express our great appreciation to editors and reviewers.

Sincerely yours,

Zuojie Zhang on behalf of the authors.